Author's response to reviews

Title: High prevalence of cardiovascular risk factors in children and adolescents with Williams-Beuren syndrome

Authors:

Daiji Takeuchi (0220teAMDaiiji@gmail.com)
Michiko Furutani (michii@imcir.twmu.ac.jp)
Yuriko Harada (ytr.koro@gmail.com)
Yoshiyuki Furutani (yoshi@imcir.twmu.ac.jp)
Kei Inai (pinai@hij.twmu.ac.jp)
Toshio Nakanishi (pnakanis@hij.twmu.ac.jp)
Rumiko Matsuoka (rumiko@imcir.twmu.ac.jp)

Version: 5 Date: 20 August 2015

Author's response to reviews:

Response to editors 20 Aug 2015

MS: 1039586234135477

Title: High prevalence of cardiovascular risk factors in children and adolescents with Williams-Beuren syndrome

BMC Pediatrics

Corresponding authors: Dr. Daiji Takeuchi and Dr. Rumiko Matsuoka

The corrected text is shown in red font in the revised manuscript.

Response to

Editor’s comment: Title Page:

Please include the email addresses of all authors on the title page. Please ensure that it is identical to the one entered on the submission system.

Response: We added email addresses of all authors on title page. Accordingly, we have revised the manuscript as follows.

Change 1: Page 1, line 22: We added “E-mail: Michiko Furutani michii@imcir.twmu.ac.jp - Yuriko Harada ytr.koro@icloud.com - Yoshiyuki Furutani yoshi@imcir.twmu.ac.jp - Kei Inai pinai@hij.twmu.ac.jp - Toshio Nakanishi pnakanis@hij.twmu.ac.jp”